Notes of the RICH1 Phone meeting 26/05/2006
People Present:
Bristol:  Richard
CERN:  Fabio, Olav, Nick 
IC:         Bill, Dave, Geoff, Trevor
Oxford: Neville
RAL:     Derek, Peter
Apologies for absence: 
CERN: Werner

IC: Gabriel
Discussed Agenda Topics:
1. Gas Enclosure 
A definite date for the installation is not yet available. The probable date is  around the 12 of June06.
Aerogel Rails: Have been tested and everything fitted well. Now the aerogel rail system is back in Milano where they will try to reduce the friction of the system. Need to conduct another test with the GE in the vertical position. The only possibility left is in the pit, but the time slot is limited to ~1 day. 
Installation hoist and pieces: All the missing bolts have been secured and all the necessary material is available. A partial assembly of the pieces has been carried out and all fits well. It is ready to be transported.
Painting: The inside of the GE has been painted by Bill and Dave using the RICH2 black paint. Two coatings were applied. All bolt holes were masked of and the o-ring machined area was not painted.
PPSPS documents: Bill and Rohan have prepared the papers and are on EDMS. Last Wednesday a presentation was given to the CERN safety coordinator. The documentation was also sent to French authorities for approval.  There have been only two questions regarding the installation. The first question; the super clamp certification was questioned because of its proposed orientation usage (upside down). The company needs to be contacted to clarify the paperwork (action Rohan).  The second question was about the rolling of the GE in the pit. The recommendation is that the rolling should not be done on a grid but on a flat plate.
Temporary Sealing of GE: Leave the quartz windows metal plates in place, use light-weight side doors (plastic) and protect the upstream and downstream windows with some plastic cover.
2. VELO Seal
The VELO Seal has been sent to CERN and should arrive at the end of next week. The 0-rings have been delivered. The installation tooling is being manufactured. The remaining components for the installation table have been ordered.  The  Seal to VELO installation will occur sometime between the 19 June06 - 9 July06. 

3. Exit Window

Two diaphragms (50% and 75%) pigment) have been sent to IC for opacity testing. Based on earlier samples, a better performance was expected from both samples but the 75% pigmented diaphragm is good enough for our requirements. The mould is out for manufacture; ready ~1-2 weeks. Once the mould is ready other diaphragms will be manufactured and tested. The estimated date for the installation of the beampipe is the 15/8/06. Last  Wednesday  a meeting took place in the pit to discuss installation of the EW,  see Bill’s report here below. 

“Mike, Derek, Peter and I visited Point 8 this morning with Rolf and Olivier to decide how to handle the RICH1 exit window installation. Here is my summary.
All agreed that an installation upstream of the magnet (from the cavern side) should be possible, using the local 1-tonne crane to deliver the exit window along a path between the TT vertical support and the magnet to a point where it can be man-handled on to a long installation bracket mounted on the RICH1 Gas Enclosure but further supported on the upper RICH1 pole extension vertical pieces. The shielding drawings are on EDMS 572793 and show two vertical M12 holes with 30mm useable depth.

Oliver Jamet says that these holes will not be covered by the TT bridge beam (which passes through upstream of the vertical shielding block). He has completed the design of the TT bridge and rail and will send us drawing as soon as they are available.

Still to be checked is the furthest downstream Z distance which can be accessed by the crane. This will also be provided by Olivier.

The brackets for the beam-pipe support wires may interfere with the crane hook access to this region but it is likely that the one on the cavern side can be removed for this operation.

Mike is happy to proceed with a design along these lines.

One thing that we did not consider this morning was the fact that this long installation bracket involves some operations above the beam-pipe. This aspect will need further thought.

Bill Cameron”

4. Spherical Mirrors
The samples from CMA have been tested: optical tests, fire test, metrology, mechanics tests and reflectivity measurement.  The next step is to expose them to fluorocarbon gas and radiation.
Fluorocarbon Exposure:

Two CMA samples (flat sample A, mirror sample 1) have been exposed to C4F10 since 19 May06. First an under-pressure was applied -700 mbar to remove air from tank and sample and then the tank was filled with C4F10 with a slight over-pressure (+100 mbar). The tank is at room temperature. The samples will be left in the tank for ~4 weeks and then will be remeasured.  The ageing can be accelerated by heating the tank.

Irradiation: The remaining two samples (flat sample B, mirror sample 2) will be brought to IONISOS at DAGNEUX near Lyon to be irradiated with a dose of 1kGy (gamma irradiation, Cobalt 60) next week (action Fabio). Small Alanine dosimeters will be placed near the samples to cross check at CERN the deposited dose.
During their visit at CMA this month, Geoff and Trevor discussed and redesigned the mirror frame system with CMA. The adjustment mechanism is based on the idea of having a simple and robust system which is locked into position and left there. The model files are available on the CMA website but there are some problems in opening the files. CMA is aware of this and it should be resolved soon. Once manufactured the rail system will be sent out to CMA so that they can test the frame with the mirrors and check repeatability. The demonstrator mirror together with a witness sample has been sent to CERN this week. 
5. Flat mirrors
All 24 flat mirrors have been coated. The alignment rig needs to be prepared (action Fabio/Carmelo).
Retainer Ring: The D0 was measured: before D0=1.71 after gluing D0=1.74. The polycarbonate pieces have been ordered.
Mechanics:  The mirror adjustment system needs to be finalized (action Geoff). The rails are designed and once manufactured they could be sent to CERN. Not yet decided where the angle adjusters will be manufactured (IC or Bristol).
6. HPD Assembly
Last week had discussions with Ken.  The layout for the patch panels is converging; an individual patch panel per column. The photon funnel is being looked at to release some space for the MDMS.  The material for the ladders is in the UK now and the machining can be scheduled. Erich and Olav will send a complete RICH2 mechanics ladder to IC.  The LV cables will have to be unplugged each time during an access.
7. Aerogel

More tiles arrived. Delivery should be completed by the end of June06.

8.  Quartz Windows
The production drawings need to be checked and then the order can be launched (action Trevor/ Mike).

9.  Magnetic Distortions

No news. There is not much room for the Syracuse solution. The backup solution is being followed by Gabriel (projector system) were the shadowing is at most ~10%.
10. Next meeting
Friday 9th of June 2006 at 10:00(GMT) 11:00(CET). 
